\begin{thebibliography}{10}

\bibitem{Sendall:03}
Sendall, S., Kozaczynski, W.:
\newblock {Model Transformation: The Heart and Soul of Model-Driven Software
  Development.}
\newblock IEEE Software \textbf{20}(5) (2003)  42--45

\bibitem{ATL:03}
B{\'e}zivin, J., Dup{\'e}, G., Jouault, F., Pitette, G., Rougui, J.E.:
\newblock {First Experiments with the ATL Model Transformation Language:
  Transforming XSLT into XQuery}.
\newblock In: {Companion of the $18^{th}$ Annual ACM SIGPLAN Conference on
  Object-Oriented Programming, Systems, Languages, and Applications, OOPSLA
  2003}, {Anaheim, CA, USA}, ACM (2003)

\bibitem{Gokhale_GTQMAP_RTAS:08}
Kavimandan, A., Gokhale, A.:
\newblock {Automated Middleware QoS Configuration Techniques using Model
  Transformations}.
\newblock In: Proceedings of the $14^{th}$ IEEE Real-Time and Embedded
  Technology and Applications Symposium (RTAS 2008), St. Louis, MO, USA (April
  2008)  93--102

\bibitem{Karsai:04}
Agrawal, A., Simon, G., Karsai, G.:
\newblock {Semantic Translation of Simulink/Stateflow Models to Hybrid Automata
  Using Graph Transformations}.
\newblock In: International Workshop on Graph Transformation and Visual
  Modeling Techniques (GT-VMT), Electronic Notes in Theoretical Computer
  Science, Barcelona, Spain, Springer-Verlag (March 2004)  43--56

\bibitem{Gokhale_Dialog_EDOC:08}
Kavimandan, A., Klemm, R., Gokhale, A.:
\newblock {Automated Context-sensitive Dialog Synthesis for Enterprise
  Workflows using Templatized Model Transformations}.
\newblock In: Proceedings of the 12th International Conference on Enterprise
  Computing (EDOC '08), Munchen, Germany, IEEE (September 2008)  159--168

\bibitem{Polack_ICMT:08}
Kolovos, D.S., Paige, R.F., Polack, F.A.C.:
\newblock {The Epsilon Transformation Language}.
\newblock In: Proceedings of the $1^{st}$ International Conference on Theory
  and Practice of Model Transformations (ICMT 2008). Volume 5063 of Lecture
  Notes in Computer Science., Zurich, Switzerland, Springer (July 2008)  46--60

\bibitem{VIATRA:02}
Csert{\'a}n, G., Huszerl, G., Majzik, I., Pap, Z., Pataricza, A., Varr{\'o},
  D.:
\newblock {VIATRA: Visual Automated Transformations for Formal Verification and
  Validation of UML Models}.
\newblock In: Proceedings of $17^{th}$ IEEE International Conference on
  Automated Software Engineering, Edinburgh, UK, IEEE (September 2002)
  267--270

\bibitem{Taentzer:04}
Taentzer, G.:
\newblock {AGG: A Graph Transformation Environment for Modeling and Validation
  of Software}.
\newblock In: International Workshop on Application of Graph Transformations
  with Industrial Relevance (AGTIVE 2003), Charlottesville, VA, USA (September
  2003)  446--453

\bibitem{GReAT:03}
Karsai, G., Agrawal, A., Shi, F., Sprinkle, J.:
\newblock {On the Use of Graph Transformations in the Formal Specification of
  Computer-Based Systems}.
\newblock In: Proceedings of IEEE TC-ECBS and IFIP10.1 Joint Workshop on Formal
  Specifications of Computer-Based Systems, Huntsville, AL, USA, IEEE (April
  2003)  19--27

\bibitem{Voelter:07}
Voelter, M., Groher, I.:
\newblock {Product Line Implementation using Aspect-Oriented and Model-Driven
  Software Development}.
\newblock In: Proceedings of the 11th Annual Software Product Line Conference
  (SPLC), Kyoto, Japan (September 2007)  233--242

\bibitem{Gerard_ISORC:08}
Thomas, F., Delatour, J., Terrier, F., G{\'e}rard, S.:
\newblock {Towards a Framework for Explicit Platform-Based Transformations}.
\newblock In: Proceedings of the $11^{th}$ IEEE International Symposium on
  Object-oriented Real-time distributed Computing (ISORC 2008), Orlando, FL,
  USA (May 2008)  211--218

\bibitem{Cuadrado_ICMT:08}
Cuadrado, J.S., Molina, J.G.:
\newblock {Approaches for Model Transformation Reuse: Factorization and
  Composition}.
\newblock In: Proceedings of the $1^{st}$ International Conference on Theory
  and Practice of Model Transformations (ICMT 2008). Volume 5063 of Lecture
  Notes in Computer Science., Zurich, Switzerland, Springer (July 2008)
  168--182

\bibitem{Clements:02}
Clements, P., Northrop, L.:
\newblock {Software Product Lines: Practices and Patterns}.
\newblock Addison-Wesley, Boston, USA (2002)

\bibitem{Jouault_ATL:08}
Jouault, F., Allilaire, F., B�zivin, J., Kurtev, I.:
\newblock {ATL: A Model Transformation Tool}.
\newblock Science of Computer Programming, Special Second Issue on Experimental
  Software and Toolkits (EST) \textbf{72}(1-2) (2008)  31--39

\bibitem{Karsai:05}
Mens, T., Gorp, P.V., Varro, D., Karsai, G.:
\newblock {Applying a Model Transformation Taxonomy to Graph Transformation
  Technology}.
\newblock In: {Proceedings of the International Workshop on Graph and Model
  Transformation (GraMoT'05)}. Volume 152 of Lecture Notes in Computer
  Science., Tallinn, Estonia, Springer-Verlag (September 2006)  143--159

\bibitem{RTCORBA:05}
{Object Management Group}:
\newblock {Real-time CORBA Specification}. {1.2} edn. (January 2005)

\bibitem{GME:01}
L\'{e}deczi, {\'A}., Bakay, {\'A}., Mar{\'o}ti, M., V{\"o}lgyesi, P.,
  Nordstrom, G., Sprinkle, J., Karsai, G.:
\newblock {Composing Domain-Specific Design Environments}.
\newblock Computer \textbf{34}(11) (2001)  44--51

\bibitem{Schurr:99}
Sch{\"u}rr, A., Winter, A.J., Z{\"u}ndorf, A.:
\newblock {PROGRES: Language and Environment}.
\newblock In Ehrig, H., Engels, G., Kreowski, H., Rozenberg, G., eds.: Handbook
  on Graph Grammars and Computing by Graph Transformation: Applications,
  Languages, and Tools, World Scientific Publishing Company (1999)  487--550

\bibitem{Salicki:01}
Salicki, S., Farcet, N.:
\newblock {Expression and Usage of the Variability in the Software Product
  Lines.}
\newblock In: The $4^{th}$ International Workshop on Software Product-Family
  Engineering. Volume 2290 of Lecture Notes in Computer Science., Bilbao,
  Spain, Springer (October 2001)  304--318

\bibitem{Gray:01}
Gray, J., Bapty, T., Neema, S.:
\newblock {Handling Crosscutting Constraints in Domain-Specific Modeling}.
\newblock Communications of the ACM (October 2001)  87--93

\end{thebibliography}

% $Id: related.tex 16757 2009-02-09 05:03:03Z gokhale $
\sqzsmall
\sqzsmall
%\sqzsmall
\section{Related Work}
\label{sec:rel}
\sqzsmall
Existing model transformation
tools~\cite{VIATRA:02,Jouault_ATL:08,Schurr:99} support some form of
HOTs. PROGRES and ATL allow specification of
type parameters while VIATRA allows development of meta transformations,
\emph{i.e.}, HOTs that can manipulate
transformation rules and hence model transformations.  Unlike MTS,
however, these tools do not provide mechanisms for separation of
variabilities from model transformations to facilitate automated
development of application families. 

A recent work synergistic to MTS appears in~\cite{Cuadrado_ICMT:08}. In
this work the authors propose (1) transformation factorization to
extract common parts of two or more transformation definitions into a
reusable, base transformation, and (2) composing transformation
definitions mapping from a single source metamodel to multiple target
metamodels, each representing a specific concern in the system being
transformed.  MTS differs from ~\cite{Cuadrado_ICMT:08} in that we focus
on composing the common (base) transformation by using the constraint
notation (as opposed to factoring out commonalities from existing
transformations), and automating the entire process of transformation
specialization (\emph{i.e.}, creating instances of transformations). 

\iffalse
% Amogh, I am commenting this out but it would be nice if we can add a
% line about QVT and where MTS could fit
The model driven architecture (MDA) development process is
centered around defining application platform-independent models and
applying (typed, and attribute augmented) transformations to these
models to obtain application platform-specific models.  In the context
of MDA, requirements and challenges in generating specialized
transformations from generic transformations are discussed
in~\cite{Kovse:02}. 
\fi

% JGG: The following could be rewritten better.

Reflective model-driven engineering (MDE)~\cite{Jouault_ATL:08}
proposes a two-dimensional MDE process by expressing model
transformations in a tool- or platform-indepe\-ndent way and
transforming expressions into actual tool- or platform-specific
model transformation expressions.  
\iffalse
There is return on investment (ROI)
associated with developing and maintaining mappings from
platform-indep\-endent transformations to platform-specific
transformations in terms of reuse, composition, customization,
maintenance etc.  The authors argue that the ROI for a two-dimensional
MDA process is greater than conventional one dimensional MDA.  
\fi
Although reflective MDE focuses on having durable transformation
expressions that naturally facilitate technological evolution and
development of tool-agnostic transformation projects, mappings still
have to be evolved with a change in platform-specific
technologies.   MTS, however, is concerned with managing and 
evolving model transformation variability in systems developed using MDE.
%Parameterization techniques supported by 
%MTS can be highly effective in managing variability of mappings from
%platform-independent to specific forms in the context of the above body
%of work. 

Asset variation points discussed in 
~\cite{Salicki:01} deal with expressing variability in models of 
product lines~\cite{Clements:02}. %,Parnas:76}.
 A variation point is identified by several 
characteristics (\emph{e.g.}, point reference, and context, use and
rationale of the variation point) that uniquely identify that point in
the product lines.  These asset variation points capture variation rules
of implementation components of a product line member. 

An aspect-oriented approach to managing transformation variability is
discussed in~\cite{Voelter:07} that relies on capturing
variability in terms of models and code generators. 
Another approach is model weaving~\cite{Gray:01}, which is used
in the composition of separate models that together define the system as
a whole.  
%Aspect models allow specifying variability that is weaved
%into a base model to form an instance of a product-line. 
Using the aspect-oriented approach requires developers to
learn a new modeling language for creating aspect models for their
product line.  In contrast, the VMM models generated by MTS use modeling
objects that are part of the source (or target) modeling languages
requiring no additional learning curve.

